Author's response to reviews

Title: Regulation of gene expression in human tendinopathy

Authors:

Scott A Jelinsky (Scott.Jelinsky@pfizer.com)
Scott A Rodeo (rodeos@hss.edu)
Jian Li (Jian.Li@pfizer.com)
Lawerence V Gulotta (GulottaL@hss.edu)
Joanne M Archambault (joarch@comcast.net)
Howard J Seeherman (Howard.Seeherman@pfizer.com)

Version: 3 Date: 25 April 2011

Author's response to reviews:

To BMC Series Editorial Production Team,

As requested, we have made a number of formatting changes to our manuscript MS:1726195765496362 regulation of gene expression in human tendinopathy as outlined below.

1. All tables are now incorporated into the main manuscript.
2. All forms of tracking have been removed.
3. Minor typographical errors have been corrected.

Sincerely,

Scott A. Jelinsky